Author’s response to reviews

Title: Respiratory mechanics in infants with severe bronchiolitis on controlled mechanical ventilation

Authors:

Pablo Cruces (pcrucesr@gmail.com)
Sebastián Gonzalez-Dambrauskas (sgdambrauskas@gmail.com)
Julio Quilodrán (julio.quilodran@gmail.com)
Jorge Valenzuela (clarkinekit@yahoo.es)
Javier Martínez (javimartinez606@gmail.com)
Natalia Rivero (natyrivero.kine@gmail.com)
Pablo Arias (pro944@gmail.com)
FRANCO DIAZ (francodiazr@gmail.com)

Version: 3 Date: 03 Jul 2017

Author’s response to reviews:

July 3rd, 2017

Editors

BMC Pulmonary Medicine

Dear Benny Yau

Thank you, the reviewers and editors for the indications for the final version of our manuscript. We added your suggestions as follows:

1. Please make sure the abstract follows the format outlined in our submission guidelines.
Please make sure the submission data is also updated with any abstract changes.
We double checked that the abstract was in accordance journal’s instructions.

2. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.
We added the conclusion section.

3. We note that table 1 contains potentially identifiable information. It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper (http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9).
Currently, the age, gender, and weight information together may compromise patient anonymity.
Please amend table 1 to address this issue, by removing the weight and age information, or providing ages as age-ranges instead.
Thank you for bringing this to our attention. Gender and weight were removed.

4. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.
5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

It will be uploaded as final clean version.

Sincerely

Franco Díaz R, M.D.
Pediatric Intensive Care Unit
Clínica Alemana de Santiago
Santiago, Chile
Pablo Cruces R, M.D.
Centro de Investigación de Medicina Veterinaria, Universidad Andres Bello.
Pediatric Intensive Critical Unit, Hospital El Carmen de Maipú.
Santiago, Chile